

RESEARCH ARTICLE

10.1029/2021EF002553

Key Points:

- The spatial pattern and temporal changes of China's vegetation community complexity were examined using over half a million field samples
- Despite China is getting greener, China's vegetation community complexity decreased during the past 30 years
- With proper vegetation conservation efforts, human activities can have a positive effect on maintaining vegetation composition complexity

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to: Q. Guo, guo.qinghua@pku.edu.cn

Citation:

Su, Y., Guo, Q., Guan, H., Hu, T., Jin, S., Wang, Z., et al. (2022). Human-climate coupled changes in vegetation community complexity of China since 1980s. *Earth's Future*, 10, e2021EF002553. https://doi. org/10.1029/2021EF002553

Received 16 NOV 2021 Accepted 6 MAY 2022

Author Contributions:

Conceptualization: Yanjun Su, Qinghua Guo Data curation: Yanjun Su, Hongcan Guan Formal analysis: Tianyu Hu Funding acquisition: Qinghua Guo Methodology: Yanjun Su, Tianyu Hu, Shichao Jin

Project Administration: Qinghua Guo Writing – review & editing: Yanjun Su, Qinghua Guo, Hongcan Guan, Tianyu Hu, Shichao Jin, Zhiheng Wang, Lingli Liu, Lin Jiang, Ke Guo, Zongqiang Xie, Shazhou An, Xuelin Chen, Zhanqing Hao, Yuanman Hu, Yongmei Huang, Mingxi Jiang, Jiaxiang Li, Zhenji Li, Xiankun Li, Xiaowei Li, Cunzhu Liang,

© 2022 The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited and is not used for commercial purposes.

Human-Climate Coupled Changes in Vegetation Community Complexity of China Since 1980s

Yanjun Su^{1,2}, Qinghua Guo^{3,4}, Hongcan Guan⁴, Tianyu Hu^{1,2}, Shichao Jin^{5,6}, Zhiheng Wang⁴, Lingli Liu^{1,2}, Lin Jiang⁷, Ke Guo^{1,2}, Zongqiang Xie^{1,2}, Shazhou An⁸, Xuelin Chen⁹, Zhanqing Hao¹⁰, Yuanman Hu¹⁰, Yongmei Huang¹¹, Mingxi Jiang¹², Jiaxiang Li¹³, Zhenji Li¹⁴, Xiankun Li¹⁵, Xiaowei Li¹⁶, Cunzhu Liang¹⁷, Renlin Liu¹⁸, Qing Liu^{19,20}, Hongwei Ni²¹, Shaolin Peng²², Zehao Shen⁴, Zhiyao Tang⁴, Xiaobo Yang²⁷, Yongchuan Yang^{28,29}, Renqing Wang²⁵, Yingzhong Xie¹⁶, Xiaoniu Xu²⁶, Xiaobo Yang²⁷, Yongchuan Yang^{28,29}, Lifei Yu³⁰, Ming Yue³¹, Feng Zhang^{32,33}, Jun Chen³⁴, and Keping Ma^{1,2}

¹State Key Laboratory of Vegetation and Environmental Change, Institute of Botany, Chinese Academy of Sciences, Beijing, China, ²University of Chinese Academy of Sciences, Beijing, China, ³Institute of Remote Sensing and Geographical Information Systems, School of Earth and Space Sciences, Peking University, Beijing, China, ⁴Institute of Ecology, College of Urban and Environmental Science, Peking University, Beijing, China, ⁵Plant Phenomics Research Centre, Nanjing Agricultural University, Nanjing, China, ⁶Jiangsu Collaborative Innovation Centre for Modern Crop Production, Nanjing Agricultural University, Nanjing, China, 7School of Biological Sciences, Georgia Institute of Technology, Atlanta, GA, USA, ⁸College of Pratacultural and Environmental Sciences, Xinjiang Agricultural University, Urumqi, China, ⁹College of Life Science, Northwest Normal University, Lanzhou, China, ¹⁰Key Laboratory of Forest Ecology and Management, Institute of Applied Ecology, Chinese Academy of Sciences, Shenyang, China, ¹¹College of Resources Science and Technology, Beijing Normal University, Beijing, China, ¹²Key Laboratory of Aquatic Botany and Watershed Ecology, Chinese Academy of Sciences, Wuhan, China, ¹³College of Forestry, Central South University of Forestry & Technology, Changsha, China, ¹⁴College of Life Sciences, Xiamen University, Xiamen, China, ¹⁵Guangxi Institute of Botany, Chinese Academy of Sciences, Guilin, China, 16 Agricultural College, Ningxia University, Yinchuan, China, 17 School of Ecology and Environment, Inner Mongolia University, Hohhot, China, ¹⁸Institute of Navel Orange, School of Life and Environmental Science, Gannan Normal University, Ganzhou, China, ¹⁹Key Laboratory of Mountain Ecological Restoration and Bioresource Utilization, Chengdu Institute of Biology, Chinese Academy of Sciences, Chengdu, China, 20 Ecological Restoration and Biodiversity Conservation Key Laboratory of Sichuan Province, Chengdu Institute of Biology, Chinese Academy of Sciences, Chengdu, China, ²¹Heilongjiang Academy of Forestry, Harbin, China, ²²State Key Laboratory of Biocontrol and Guangdong Key Laboratory of Plant Resources, School of Life Sciences, Sun Yat-Sen University, Guangzhou, China, ²³School of Life Sciences, Nanjing University, Nanjing, China, 24Zhejiang Tiantong Forest Ecosystem National Observation and Research Station, School of Ecological and Environmental Sciences, East China Normal University, Shanghai, China, ²⁵School of Life Sciences, Shandong University, Qingdao, China, ²⁶Department of Forest Science, Anhui Agricultural University, Hefei, China, 27The College of Ecology and Environment, Hainan University, Haikou, China, 28Key Laboratory of Three Gorges Reservoir Region's Eco-Environment, Ministry of Education, Chongqing University, Chongqing, China, ²⁹National Centre for International Research of Low-Carbon and Green Buildings, Chongqing University, Chongqing, China, ³⁰College of Life Science, Guizhou University, Guiyang, China, ³¹Key Laboratory of Resource Biology and Biotechnology in Western China, Northwest University, Xi'an, China, 32Institute of Loess Plateau, Shanxi University, Taiyuan Shanxi, China, 33School of Life Science, Shanxi University, Taiyuan, China, 34National Geomatics Center of China, Beijing, China

Abstract Vegetation community complexity is a critical factor influencing terrestrial ecosystem stability. China, the country leading the world in vegetation greening resulting from human activities, has experienced dramatic changes in vegetation community composition during the past 30 years. However, how China's vegetation community complexity and temporally remains unclear. Here, we examined the spatial pattern of China's vegetation community complexity and its temporal changes from the 1980s to 2015 using two vegetation maps of China as well as more than half a million field samples. Spatially, China's vegetation community distribution is primarily dominated by elevation, although temperature and precipitation can be locally more influential than elevation when they become the factors limiting plant growth. Temporally, China's vegetation greening trend. Prevailing climate warming across China exhibits a significant negative correlation with the decrease in vegetation community complexity, but this correlation varies with biogeographical regions. The intensity of human activities have an overall negative influence on vegetation community complexity, but vegetation and restoration efforts can have a positive



Renlin Liu, Qing Liu, Hongwei Ni, Shaolin Peng, Zehao Shen, Zhiyao Tang, Xingjun Tian, Xihua Wang, Renqing Wang, Yingzhong Xie, Xiaoniu Xu, Xiaobo Yang, Yongchuan Yang, Lifei Yu, Ming Yue, Feng Zhang, Jun Chen, Keping Ma effect on maintaining vegetation composition complexity, informing the critical role of vegetation management policies in achieving the sustainable development goal.

Plain Language Summary China, as a country leading in vegetation greening, has experienced remarkable changes in vegetation community compositions. However, its vegetation community complexity, as an important factor indicating terrestrial ecosystem stability and sustainability, has been rarely studied. This study provides a comprehensive understanding on the spatial and temporal changes in China's vegetation community complexity from the1980s to 2015, and highlights a significant decreasing trend despite observable vegetation greening. Climate warming has an overall significant negative correlation with the decrease in vegetation community complexity, but their correlation varies with biogeographical regions. The intensity of human activities have an overall negative influence on vegetation community complexity, but vegetation conservation and restoration efforts can exhibit a positive effect on maintaining vegetation composition complexity, informing the critical role of vegetation management policies in achieving the sustainable development goal.

1. Introduction

Vegetation community is a basic geographic unit with relatively uniform plant species compositions (Jorgensen & Fath, 2014). The heterogeneity of vegetation community types within a terrestrial ecosystem, that is, vegetation community complexity, can mediate its biological processes (e.g., energy harnessing, water and nutrient uptake and cycling) (Bonan, 2008; Davin & Noblet-Ducoudré, 2010; Duveiller et al., 2018; Goodman, 1975), and therefore influences its stability and sustainability to environmental changes (Fornoff et al., 2019; Geng et al., 2019; Goodman, 1975; Ives & Carpenter, 2007; Pennekamp et al., 2018; Pimm, 1984; Schnabel et al., 2019). Revealing factors contributing to vegetation community complexity changes is key to understanding terrestrial ecosystem processes and formulating vegetation management policies (Bryan et al., 2018).

Human actives and global climate change are believed to be two major factors influencing vegetation community complexity changes (Fang et al., 2018; Komatsu et al., 2019), which can both lead to either losses or gains in vegetation community complexity. For example, vegetation community complexity may increase in areas with vegetation restoration and conservation efforts (Bryan et al., 2018; Y. Li et al., 2018; Piao et al., 2015; Strassburg et al., 2017), while decrease in areas with resource overexploitation (e.g., mining, farming, and grazing activities, which may eliminate vegetation coverage or alter vegetation community type) (Fang et al., 2018; Hou et al., 2019; Tian et al., 2014; Y. Wang et al., 2020). Similarly, global climate change may increase vegetation community complexity in areas with extended growing season length (Fang et al., 2004; Fridley et al., 2016; Jeong et al., 2011), while reduce it in areas with an increasing frequency of extreme climate events (e.g., droughts, floods) (Bronstert, 2003; Fettig et al., 2019; Sankaran, 2019).

China, as a country with the world's sixth-largest vegetated area (Chen et al., 2019), has undergone dramatic changes in vegetation distribution and composition during the past 30 years (Su et al., 2020; Zhu et al., 2016). It is reported that China alone accounts for 25% of global anthropogenic vegetation greening since 2000 (including both cultivated greening and reforestation) (Chen et al., 2019), and over 40% of China's vegetated areas have experienced changes in vegetation type due to intensive human activities and unprecedented climate change (Fang et al., 2004; Gao et al., 2019; Su et al., 2020). These changes may alter the complexity of vegetation communities in China's terrestrial ecosystems, and influence their capability of sequestering atmospheric CO_2 to mitigate global warming (Fang et al., 2018; Huang et al., 2018; X. Tang et al., 2018). However, it is still unclear how China's vegetation community complexity changed over the past three decades under the coupled influence of human activities and global climate change, which needs to be investigated urgently.

Spatially explicit vegetation maps that provide information at the formation/subformation levels are the foundation for quantifying vegetation community complexity and examining spatial and temporal variations through time. Recent advances in remote sensing and crowdsourcing techniques have enabled researchers to map national scale vegetation types with high accuracy and efficiency (Su et al., 2016, 2020). We updated the vegetation map of China for the year 2015 using a strategy of "crowdsourcing-change detection-classification-expert knowledge" (Su et al., 2020). By comparing it with the original vegetation map of China from the 1980s (X. Zhang





Figure 1. Distribution of field data used in this study. Both point-based field samples (plot and crowdsourced samples) and polygon-based forest resources survey data were collected. Each gray square cell in the background (100×100 km in area) represents a unit area used for calculating vegetation community complexity.

et al., 2007), we are able, for the first time, to investigate spatial patterns and temporal changes of China's vegetation community complexity.

Specifically, here we aim to address the following questions. How is vegetation community complexity spatially distributed across China, and what factors influence its spatial distribution? How has China's vegetation community complexity changed from the 1980s to 2015, and how do human activities and changes in climate contribute to changes in vegetation community complexity? A total of 523,398 field samples across China and 547,000 km² forest resources survey data were collected to quantify and validate vegetation community complexity along with the two vegetation maps of China. Two terrain variables (terrain elevation and terrain roughness), two climate variables (mean annual temperature [MAT] and mean annual precipitation [MAP]), two anthropogenic variables (gross domestic product [GDP] and human modification index [HMI]), and the annual maximum-value composite (MVC) of normalized difference vegetation index (NDVI) were derived to analyze factors influencing the spatial and temporal variations of China's vegetation community complexity during the past 30 years.

2. Materials and Methods

2.1. Field Data

We collected two types of field data: point-based field samples and polygon-based forest resources survey data to evaluate the accuracy of vegetation community complexity estimates from vegetation maps (Figure 1). A total of 523,398 point-based samples were collected from three sources, including (a) 228,156 crowdsourced samples taken since 2018 using LiVegetation (a self-developed mobile phone application designed to collect

Table	1
rable	T.

A Summary of the Ancillary Datasets Used in This Study, Including Dataset Name, Year, Spatial Resolution, Derived Factors, and Source

Dataset	Year	Resolution	Factors	Source	
Biogeographical region map of China	N/A	Vector	N/A	(Z. Wang et al., 2012)	
Corrected Shuttle Radar Topography Mission product	2000	30 m	Terrain elevation, terrain roughness	(Zhao et al., 2018)	
Monthly average temperature and monthly total precipitation	1985–2015	0.5°	Mean annual temperature, mean annual precipitation	(Legates & Willmott, 1990a; Legates & Willmott, 1990b; Willmott, 2001)	
Global Inventory Monitoring and Modeling System	1985–2015	1/12°	Annual maximum-value composite of normalized difference vegetation index	(Pinzon & Tucker, 2014; Tucker et al., 2005)	
Gross domestic production	2015	1 km	Gross domestic production	(Han et al., 2011)	
Global human modification index	2015	1 km	Human modification index	(Kennedy et al., 2019)	

Note. N/A represents the corresponding information is not available

crowdsourced field samples for producing vegetation maps) (Jin et al., 2021; Su et al., 2020), (b) 170,172 crowdsourced samples from the Chinese Field Herbarium (CFH) platform collected since 2014 (C. Li, 2018), and (c) 125,070 vegetation plot samples taken since 2010. The two crowdsourced samples recorded the location, sampling time, and vegetation formation/subformation type, and vegetation plot samples recorded all abovementioned information as well as field inventory data. Each point-based sample was manually examined based on the following criteria. They should (a) include vegetation formation/subformation information that can be matched to areas on vegetation maps; and (b) contain a latitude and longitude record accurate to at least three decimal places (approximately equivalent to a 100 m georeferencing accuracy). Samples that did not satisfy these criteria were excluded. A total of 230,401 samples were retained for the following analyses. Because only natural vegetation was considered in the calculation of vegetation community complexity, samples with a vegetation group type of cultural vegetation (food and cash crops) were further excluded in the following analyses.

Polygon-based forest resources survey data were distributed in northeast China and Hubei Province across a total area of around 547,000 km² (Figure 1). These data, collected since 2015, were a part of the national forest resources survey program and were provided by the National Forestry and Grassland Administration. Each polygon represented a forest stand and provided information about the dominant tree species, which was then matched with the appropriate vegetation formation/subformation type of China's vegetation maps (Su et al., 2020).

2.2. Vegetation Maps of China

In this study, two vegetation maps were used as the foundation to calculate the vegetation community complexity of China and quantify its spatial and temporal changes. The original vegetation map of China (1:1,000,000) (hereafter the original vegetation map) was produced by a team of over 250 scientists and delineated the distribution of 866 vegetation formation/subformation types (which could be grouped into 55 vegetation types/subtypes, and 12 vegetation group types) (X. Zhang et al., 2007). The original vegetation map entirely relied on field surveys, which started in the early 1980s and lasted around 10 years. The updated vegetation map of China (1:1,000,000) (hereafter the updated vegetation map) was produced as an update to the original vegetation map using a strategy called "crowdsourcing-change detection-classification-expert knowledge" (Su et al., 2020). The updated vegetation map used 2015 as the baseline year and utilized the same vegetation formation/subformation classification system as the original vegetation map. Detailed information on the procedure for producing the updated vegetation map can be found in Su et al. (2020). Similar to field samples, cultural vegetation was excluded from both the original and updated vegetation maps since only natural vegetation was considered in the vegetation community complexity calculation.

2.3. Ancillary Datasets

Six ancillary datasets were assembled to explore factors influencing spatial and temporal variations in China's vegetation community complexity, including China's biogeographical region map, two terrain surfaces, climate surfaces, NDVI, GDP, and HMI (Table 1). The biogeographical region map of China was provided by Z. Wang

et al. (2012), which was a revision of the map published by the Editorial Board for Physical Geography of China (Wu, 1985). This map divided China into seven biogeographical regions: southeast China, the eastern Himalayas, the Tibetan Plateau, north China, the Mongolian Plateau, northeast China, and northwest China, represented by Roman numerals from I to VII, respectively.

Two terrain surfaces, including a digital elevation model (DEM) and terrain roughness, were used in this study. The national DEM of China at 30-m resolution was derived from the corrected Shuttle Radar Topography Mission (SRTM) product (Table 1), which corrected the systematic overestimation bias of the original SRTM product in vegetated areas (Zhao et al., 2018). Terrain roughness was derived from the SRTM DEM and calculated as the standard deviation of elevation in a unit area (100×100 km in this study).

Two types of climate surfaces, monthly average temperature (Legates & Willmott, 1990a; Willmott, 2001) and monthly total precipitation (Legates & Willmott, 1990b; Willmott, 2001), were collected for the time period of 1985–2015 to examine trends in climate conditions of China (Table 1). These data were interpolated from a large number of weather station records at a spatial resolution of 0.5° latitude $\times 0.5^{\circ}$ longitude (Legates & Willmott, 1990a, 1990b; Willmott, 2001). We further derived MAT (calculated as the mean of monthly average temperatures of each year) and MAP (calculated as the sum of monthly total precipitations of each year) from the monthly temperature and precipitation data, which were used in the following analyses (Table 1).

NDVI is a remotely sensed parameter that can represent vegetation greenness and physiological status (Gan et al., 2021; Ju & Masek, 2016; Piao et al., 2003). Here, we used the most recent time-series NDVI product (NDVI3g) from the Global Inventory Monitoring and Modeling System (GIMMS) project to evaluate vegetation greenness dynamics during last three decades (Table 1) (Pinzon & Tucker, 2014; Tucker et al., 2005). The GIMMS NDVI product was produced from Advanced Very-High-Resolution Radiometer data at a spatial resolution of $1/12^{\circ}$ latitude $\times 1/12^{\circ}$ longitude and a temporal frequency of a half month (Pinzon & Tucker, 2014). In this study, we collected GIMMS NDVI data from 1985 to 2015, and calculated a MVC of NDVI for each year to reduce the influence of cloud contamination and atmospheric attenuation (Table 1) (Su et al., 2017; van Leeuwen et al., 1999).

In addition, the national GDP and HMI were collected to reflect the intensity of human activities and evaluate the influence of human activities on vegetation community complexity changes. Here, a gridded national GDP distribution product from Han et al. (2011) was used (Table 1), which was interpolated from 2015 county-level GDP data created by the National Bureau of Statistics of China using land cover type, nighttime light remote-sensing data, and population density as covariables. HMI is a cumulative measure of human modifications on terrestrial areas by examining 13 anthropogenic stressors covering human settlement, agriculture, transportation, mining and energy production, and electrical infrastructure. In this study, we used the HMI product of the year 2015 from Kennedy et al. (2019), which was provided at 1-km resolution with values ranging from 0 to 1, with 1 representing the largest proportion of a landscape being modified by human activities (Table 1). Because the influence of human activities on vegetation community complexity might be a cumulative effect, especially vegetation conservation and restoration activities, here we used the 2015 GDP and HMI instead of its magnitude of changes during the past 30 years.

As can be seen in Table 1, one of the major limitations of the collected datasets was that they were in different spatial resolutions (except the biogeographical map of China) (Table 1). To be consistent with the calculation procedure of vegetation community complexity, all these datasets were resampled to a spatial resolution of 100 km. Considering the fact that the spatial resolutions of these datasets were all finer than 100 km, the downsampling procedure should not have a significant influence on the following analyses.

2.4. Calculation of Vegetation Community Complexity

In this study, vegetation community complexity was quantified by Shannon entropy, a commonly used index for indicating the complexity of an ecosystem or landscape (Nagendra, 2002; Peet, 1975; Peters et al., 2019) by examining the appearance probability of each type of individuals composed of the ecosystem or landscape (Shannon, 1948). Here, we calculated the vegetation community complexity, that is, the Shannon entropy, using the following equation,



$$VCC = -\sum_{i=1}^{S} p_i \ln p_i \tag{1}$$

where VCC is the vegetation community complexity, S is the total number of vegetation formation/subformation types in a unit area (defined as a 100×100 km cell in Figure 1), and p_i is the probability of appearance of a vegetation formation/subformation in a unit area.

To calculate vegetation community complexity from vegetation maps, the probability of appearance of a vegetation formation/subformation in a unit area was calculated using the following equation,

$$p_i = \frac{A_i}{A} \tag{2}$$

where A_i is the total area of a vegetation formation/subformation in a unit area and A is the area of the unit. Vegetation community complexity estimates were calculated for both the original and updated vegetation maps. The estimate from the original vegetation map was used to represent the spatial pattern of China's vegetation community complexity since it was less disturbed by recent human activities. The difference between estimates from the updated and original vegetation maps was calculated to represent vegetation community complexity change from the 1980s to 2015. Note that units with an area smaller than 1 km² were excluded in the calculation of vegetation community complexity.

Moreover, vegetation community complexity from point-based field samples and polygon-based forest resources survey data were also calculated, which were used to evaluate the accuracy of those derived from the updated vegetation map. For point-based field samples, the probability of appearance of a vegetation formation/subformation in a unit area was calculated using the following equation,

Ì

$$p_i = \frac{n_i}{N} \tag{3}$$

where n_i is the total number of field samples belonging to a vegetation formation/subformation in a unit area, and N is the total number of field samples in the unit area. For polygon-based forest resources survey data, the probability of appearance of a vegetation formation/subformation in a unit area was calculated using the same equation as that for vegetation maps. Since the forest resources survey data did not include vegetation types beyond forest, the resulting vegetation community complexity might be underestimated.

2.5. Statistical Analyses

The accuracy of vegetation community complexity estimates from vegetation maps was evaluated against those from point-based field samples and polygon-based forest resources survey data. To ensure the representativeness of the field data estimates, we excluded cells with fewer than 2,000 field samples and cells with less than 60% coverage of forest resources survey data. Statistical accuracy was assessed using both coefficient of determination (R^2) and the *p*-value of each statistical test.

To analyze the factors contributing to spatial and temporal variations in the vegetation community complexity, correlations between the original vegetation community complexity and geolocation, terrain elevation, terrain roughness, MAT, and MAP were evaluated using linear regression analysis for terrestrial ecosystems of entire China as well as those of each biogeographical region. Since terrain elevation, terrain roughness, MAT, and MAP had strong influences on the spatial distribution of China's vegetation community complexity, and terrain elevation and roughness had significant influences on MAT and MAP, we further applied path analysis to describe the directed dependencies among them (Heise, 1975). All variables were normalized between 0 and 1 before being fed into the path analysis model. The path analysis was conducted using the Amos software, and the significance of each path was evaluated at a significance level of 0.05.

The significance of the change in China's vegetation community complexity was assessed using the paired two-tailed Student *t*-test under the null hypothesis that there were no significant differences between the vegetation community complexity estimates from the original and updated vegetation maps. Three statistical measures were reported in this study: *t*-value, degree of freedom (DOF), and *p*-value. If the *p*-value was less than 0.05, the



Figure 2. Accuracy assessment of vegetation community complexity (VCC) estimates using vegetation maps. Scatter plots between vegetation community complexity estimates from the updated vegetation map and (a) those from field samples and (b) those from forest resources survey data. Red dashed lines are fitted lines.

null hypothesis was rejected, indicating there was a significant change in China's vegetation community complexity; otherwise, there was no significant change. Temporal trends in MAT, MAP, and NDVI were analyzed using linear regression analysis on each cell. If a cell had a significant linear correlation between a given factor and time (p-value < 0.05), it was recognized as having a significant temporal trend for the corresponding factor. The magnitude of change for the corresponding factor in the cell was calculated as the slope of the linear fit multiplied by the length of the time period. The contribution of climate conditions and human activities to the spatial and temporal changes in China's vegetation community complexity were evaluated using linear regression analysis. Three statistical measures, including slope, Pearson's correlation coefficient, and p-value, were reported. These analyses were conducted for terrestrial ecosystems of entire China and those of each biogeographical region.

3. Results

In this study, both in situ vegetation community complexity estimates (i.e., from field samples and from forest resources survey data) showed good agreements with estimates derived from vegetation maps ($R^2 \ge 0.45$) (Figure 2), indicating vegetation maps are reliable sources for evaluating spatial and temporal variations in China's vegetation community complexity. To explore the spatial patterns and driving factors, we used the vegetation community complexity distribution derived from the original vegetation map of China. The resulting vegetation community complexity reveals large geographical variations across China following a normal distribution (Figures 3a and 3b). There is a weak spatial pattern of increasing complexity from north to south and from east to west (Figure 3a and Figure S1 in Supporting Information S1). Specifically, the eastern Himalayas had the highest average vegetation community complexity, followed by northwest China, the Mongolian Plateau, the Tibetan Plateau, northeast China, southeast China, and north China (Figures 3a and 3c). The vegetation community complexity regions (Figure 3c). Spatial variation in vegetation community complexity was dominated by variations in terrain, which shows a strong hump-shaped relationship with elevation and a strong linear relationship with terrain roughness (Figures 3d and 3e).

The influence of climate conditions on vegetation community complexity varied by biogeographical regions, although there was an overall decrease in vegetation community complexity as both MAT (Pearson's correlation coefficient (r) = -0.25, p < 0.05) and MAP (r = -0.12, p < 0.05) increased (Figure 4 and Table S1 in Supporting Information S1). MAT had a negative correlation with vegetation community complexity in all biogeographical regions, with the exception of the Tibetan Plateau and northeast China (Figure 4a). This negative correlation was the strongest in north China (r = -0.78, p < 0.05), followed by northwest China (r = -0.58, p < 0.05), southeast China (r = -0.46, p < 0.001), the eastern Himalayas (r = -0.41, p < 0.05), and the Mongolian Plateau (r = -0.28, p < 0.05) (Table S1 in Supporting Information S1). The Tibetan Plateau was the only biogeographical region to show a positive correlation between vegetation community complexity and MAT (r = 0.33, p < 0.05), while northeast China was the only biogeographical region to have no significant correlation between vegetation complexity and MAT (Figure 4a and Table S1 in Supporting Information S1). Correlations for MAP were weaker than those for MAT across China. Three of the seven biogeographical regions (southeast China, the Mongolian Plateau, and northeast China) showed no significant correlation between MAP and vegetation community





Figure 3.



Figure 4. Correlations between vegetation community complexity and climate. Scatter plots between vegetation community complexity and (a) average mean annual temperature (MAT) and (b) average mean annual precipitation (MAP) from 1985 to 2015. All represents data points from all biogeographical regions. Red dashed lines are fitted lines and are presented only if there were significant correlations (p < 0.05).

complexity (Figure 4b). As to the remaining four biogeographical regions, vegetation community complexity was negatively correlated with MAP in the eastern Himalayas (r = -0.38, p < 0.05) and north China (r = -0.33, p < 0.05), and was positively correlated with MAP in the Tibetan Plateau (r = 0.16, p < 0.05) and northwest China (r = 0.50, p < 0.05) (Figure 4b and Table S1 in Supporting Information S1).

Regarding temporal changes, we quantified the change in vegetation community complexity as the difference between estimates from the updated (2015) and original (1980s) vegetation maps. Overall, China's vegetation community complexity decreased slightly (mean = -0.07, p < 0.05) during the past 30 years, albeit with large spatial discrepancies (Figures 5a and 5b, and Table S2 in Supporting Information S1). Areas with a negative change in vegetation community complexity for 60% of the cells was less than 0.1 (Figure 5a). Nearly 81% of areas having a substantial negative change (<-0.3) occurred in southeast China, the eastern Himalayas, and northwest China. Southeast China alone accounted for half of these changes (Figures 5a and 5c). Northeast China also showed a decreasing trend (Table S2 in Supporting Information S1), but negative changes were largely moderate (>-0.3) (Figures 5a and 5c). Areas connecting the Mongolian Plateau and north China accounted for 34% of the cells with a positive change greater than 0.1 (Figure 5a), although the overall trends for these two biogeographical regions were insignificant (Table S2 in Supporting Information S1). The remaining cells, those with a positive change larger than 0.1, were primarily scattered across southeast China and the Tibetan Plateau (Figure 5a).

Both climatic and anthropogenic factors showed significant correlations with vegetation community complexity change, with the exception of MAP (Figure 6 and Figure S3a in Supporting Information S1). A large proportion of China experienced a significant increase in MAT (Figure S4a in Supporting Information S1), which had a positive effect on vegetation community complexity (r = 0.15, p < 0.001) (Figure 6a and Table S3 in Supporting Information S1). However, this positive correlation for changes between MAT and vegetation community existed only in southeast China, the Tibetan Plateau, and north China (Figure 6a). In general, human activities had a negative influence on vegetation community complexity, decreasing with both GDP (r = -0.16, p < 0.05) and HMI (r = -0.13, p < 0.05) (Figures 6b and 6c and Table S3 in Supporting Information S1). Negative correlations with both GDP and HMI were the strongest in northwest China (GDP: r = -0.32, p < 0.05; HMI: r = -0.33, p < 0.05), followed by northeast China (GDP: r = -0.29, p < 0.05; HMI: r = -0.23, p

Figure 3. Spatial pattern of China's vegetation community complexity. (a) Vegetation community complexity distribution derived from the original 1980s vegetation map of China. Roman numerals I–VII represent the biogeographical regions identified as southeast China, the eastern Himalayas, the Tibetan Plateau, north China, the Mongolian Plateau, northeast China, and northwest China. The Roman numerals representing each biogeographical region will be used in remaining graphs. (b) Histogram of vegetation community complexity, where μ and σ represent mean and standard deviation, respectively. (c) Bar chart of vegetation community complexity (mean + standard deviation) for the seven biogeographical regions. (d) Scatter plot between vegetation community complexity and elevation. (e) Scatter plot between vegetation community complexity and terrain roughness.





Figure 5. Changes in China's vegetation community complexity from the 1980s to 2015. (a) Map of changes in vegetation community complexity, calculated as the difference between that derived from the updated vegetation map and that derived from the original vegetation map. (b) Histogram of vegetation community complexity change for all regions, where μ and σ represent mean and standard deviation, respectively. (c) Boxplot of vegetation community complexity change for the seven biogeographical regions.

HMI, the Tibetan Plateau showed a negative correlation (r = -0.17, p < 0.05), while the Mongolian Plateau showed no significant correlation (Figure 6c), and interestingly southeast China showed a positive correlation (r = 0.19, p < 0.05) (Figure 6c and Table S3 in Supporting Information S1).

4. Discussion

Overall, spatial patterns of China's vegetation community complexity follow the distribution of mountains (Figures 3a and 3e, and Figure S4 in Supporting Information S1). Large variations in local climate brought by steep elevation changes in mountainous areas provide diverse habitats for plant communities (Z. Tang et al., 2006; J.-T. Zhang et al., 2016), where richer vegetation formation/subformation types can develop. The general negative correlations of vegetation community complexity with MAT and MAP may also reflect the controlling effect of elevation since both MAT and MAP have a significant negative correlation with elevation (Figure S4 in Supporting Information S1 and Figure 5). However, climate conditions can outweigh local elevation influences when they become the factors limiting plant growth (Figure 4). For example, the Tibetan Plateau has low MAT and





Change of community complexity

Figure 6. Factors contributing to changes in China's vegetation community complexity. Scatter plots between vegetation community complexity change and (a) MAT change from 1985 to 2015 (Δ MAT), (b) 2015 gross domestic productivity (GDP, natural logarithm transformed), and (c) 2015 human modification index (HMI). All includes data points from all biogeographical regions. Red dashed lines are fitted lines and are presented only if there were significant correlations (p < 0.05).

MAP due to its high elevation and the blocking effect of the Hengduan and Himalayan Mountains to the east and south Asian monsoons (Stokstad, 2020; D. Zhang et al., 2000), preventing the formation of complex vegetation communities (Figure 3a). As a result, increases in MAT and MAP in this region may have a more prominent effect on increasing vegetation community complexity (Figure 4). In northwest China, precipitation is the major limiting factor for plant growth (mean MAP = 126 mm) (X. Li et al., 2011). Increases in MAP can support the formation of more complex vegetation communities (Du et al., 2015; Y. Liu et al., 2018), correspondingly raising the vegetation community complexity (Figure 4b). The dominated vegetation types of a region also have influence on the spatial distribution of vegetation community complexity cells are mainly contributed by woody plants, while the high vegetation community complexity cells in the Mongolian Plateau, the Tibetan Plateau, and northwest China are mainly contributed by herbaceous plants (Figure S6 in Supporting Information S1). Herbaceous plants have a generally higher vegetation community complexity than woody plants (Lu et al., 2018; Su et al., 2020), which may lead to the relatively high vegetation community complexity in northwest China and the Mongolian Plateau (Figure 3 and Figure S6 in Supporting Information S1).

A vegetation greening trend (increasing NDVI through time) in areas beyond northwest China during the past 30 years was observed in this study (Figure S3c in Supporting Information S1), which is similar to findings in previous studies (Chen et al., 2019; Piao et al., 2015; Zhu et al., 2016). However, vegetation greening does not necessarily correspond to an increase in vegetation community complexity (Figure S2b in Supporting Information S1). Instead, a significant decreasing trend in vegetation community complexity was observed in this study (Figures 5a and 5b and Table S2 in Supporting Information S1). Prevailing climate warming is also an important factor mediating vegetation community complexity (Figure 6a and Figure S3a in Supporting Information S1). In areas with high elevation, temperature is a major factor limiting plant growth (Barber et al., 2000; Körner & Paulsen, 2004), and increasing MAT may provide a more conducive environment for plant growth (e.g., increased vegetation coverage in grasslands and alpine ecosystems in the Tibetan Plateau and northwest China) (Figure S4c in Supporting Information S1), which therefore may increase vegetation community complexity or reduce vegetation community complexity loss (Figure 5a). On the other hand, the contribution of MAP to vegetation community complexity was insignificant and might be caused by the fact that very few cells were observed with significant trends in MAP over the past 30 years (Figure S2a in Supporting Information S1 and Figure 3b). Nevertheless, with the increased frequency of extreme weather events (e.g., drought, flood) being reported (Cai et al., 2014; Coumou & Rahmstorf, 2012), we should not overlook its potential influence on vegetation community complexity.

Human activities can alter vegetation distribution and composition through deforestation, grazing, cultivation, mining, etc. (Figure S7 in Supporting Information S1). With the aim of increasing GDP and improving quality of life, such activities generally have negative influences on vegetation community complexity (Figures 6b and 6c). For example, in the eastern Himalayas, net losses in forest and grassland cover were observed (Figure S7a and c in Supporting Information S1), possibly caused by deforestation and overgrazing; on the Mongolian Plateau, net losses in grasslands were observed (Figure S7c in Supporting Information S1), possibly the result of overgrazing and mining (Tao et al., 2015; Z. Wang et al., 2017); and in northeast China, net losses in forest and shrub were observed (Figure S7a and b in Supporting Information S1), possibly caused by the expansion of cultural lands (Figure S7d in Supporting Information S1). Moreover, in the eastern Himalayas and Mongolian Plateau, widespread shrub encroachments were also observed (Figure S7b in Supporting Information S1), which might be a coupled effect from human activities (losses in grassland and forest) and climate warming (D'Odorico et al., 2010; Thomson et al., 2018). All of these changes tend to simplify vegetation community composition, and therefore reduce vegetation community complexity (Figure S8 and Table S4 in Supporting Information S1).

However, vegetation conservation and restoration efforts with proper polices can maintain or even increase vegetation community complexity. China has made aggressive efforts to rehabilitate vegetation and combat desertification in the Mongolian Plateau, north China, and south China, through projects such as the Three-North Shelter Forest Program and Grain for Green Project (J. Liu et al., 2008). In the Mongolian Plateau and north China, such efforts have significantly decreased the cumulative area associated with cultural vegetation and increased forested areas (Figure S7a and d in Supporting Information S1), leading them to have some of the most clustered areas showing significant increases in vegetation community complexity (Figure 5a). Human activities have a more profound influence on vegetation community complexity in southeast China. Two of China's economic centers (Yangtze Delta and Guangdong Province) have led a rapid expansion of GDP in this region (Hu & Wang, 2006), which comes partially at the cost of overexploitation of vegetation resources, thereby reducing overall vegetation community complexity (Figure 5a and Table S2 in Supporting Information S1) (Du et al., 2019; Seto et al., 2012). Meanwhile, in these economic centers, the importance of vegetation conservation was recognized early on, and vegetation composition has remained relatively stable (Figure S7 in Supporting Information S1). This recognition might explain how HMI significantly reduces potential decreases in vegetation community complexity (Figure 6c). Recently, afforestation efforts have begun in other areas of the region (Figure S7a in Supporting Information S1), but the uniform tree species compositions used have not yet improved vegetation community complexity (Figure S8a and Table S4 in Supporting Information S1) (Cao et al., 2011; Chazdon, 2008; Lautenbach et al., 2017).

5. Conclusions

In summary, this study reveals spatial patterns in China's vegetation composition complexity, and highlights a significant decreasing trend in complexity during the past 30 years despite observable vegetation greening. Both climate warming and human activity intensity can mediate changes to vegetation composition complexity, and our findings suggest that properly designed vegetation conservation and restoration efforts can have a positive effect on maintaining vegetation composition complexity. Nevertheless, there is still debate regarding the complexity–stability relationship (Allesina & Tang, 2012; Goodman, 1975; Grilli et al., 2017; Ives & Carpenter, 2007; Loreau & Mazancourt, 2013; McCann, 2000; Mougi & Kondoh, 2012; Pimm, 1984). How changes in vegetation composition composition complexity affect the resilience of terrestrial ecosystems to changing climate, and therefore the global carbon cycle, needs further investigations. Moreover, we still lack a complete understanding how plant species composition changes (e.g., native vs. invasive plant species) may influence vegetation community complexity, which needs further investigations as well.

Data Availability Statement

The data used to analyze the spatial pattern and temporal changes of China's vegetation community complexity is accessible here: https://doi.org/10.6084/m9.figshare.14511630.

Acknowledgments

This study was supported by the Strategic Priority Research Program of Chinese Academy of Sciences (XDA19050401), the Second Tibetan Plateau Scientific Expedition and Research Program (STEP) (2019QZKK0301), and the Information Plan of Chinese Academy of Sciences (CAS-WX2021SF-0112).

References

Allesina, S., & Tang, S. (2012). Stability criteria for complex ecosystems. *Nature*, 483(7388), 205–208. https://doi.org/10.1038/nature10832
Barber, V. A., Juday, G. P., & Finney, B. P. (2000). Reduced growth of Alaskan white spruce in the twentieth century from temperature-induced drought stress. *Nature*, 405(6787), 668–673. https://doi.org/10.1038/35015049

- Bonan, G. B. (2008). Forests and climate change: Forcings, feedbacks, and the climate benefits of forests. *Science*, 320(5882), 1444–1449. https://doi.org/10.1126/science.1155121
- Bronstert, A. (2003). Floods and climate change: Interactions and impacts. *Risk Analysis*, 23(3), 545–557. https://doi.org/10.1111/1539-6924.00335
 Bryan, B. A., Gao, L., Ye, Y., Sun, X., Connor, J. D., Crossman, N. D., et al. (2018). China's response to a national land-system sustainability emergency. *Nature*, 559(7713), 193–204. https://doi.org/10.1038/s41586-018-0280-2
- Cai, W., Borlace, S., Lengaigne, M., van Rensch, P., Collins, M., Vecchi, G., et al. (2014). Increasing frequency of extreme El Niño events due to greenhouse warming. *Nature Climate Change*, 4(2), 111–116. https://doi.org/10.1038/nclimate2100
- Cao, S., Chen, L., Shankman, D., Wang, C., Wang, X., & Zhang, H. (2011). Excessive reliance on afforestation in China's arid and semi-arid regions: Lessons in ecological restoration. *Earth-Science Reviews*, 104(4), 240–245. https://doi.org/10.1016/j.earscirev.2010.11.002
- Chazdon, R. L. (2008). Beyond deforestation: Restoring forests and ecosystem services on degraded lands. *Science*, 320(5882), 1458–1460. https://doi.org/10.1126/science.1155365
- Chen, C., Park, T., Wang, X., Piao, S., Xu, B., Chaturvedi, R. K., et al. (2019). China and India lead in greening of the world through land-use management. *Nature Sustainability*, 2(2), 122–129. https://doi.org/10.1038/s41893-019-0220-7
- Coumou, D., & Rahmstorf, S. (2012). A decade of weather extremes. *Nature Climate Change*, 2(7), 491–496. https://doi.org/10.1038/nclimate1452
 Davin, E. L., & Noblet-Ducoudré, N. d. (2010). Climatic impact of global-scale deforestation: Radiative versus nonradiative processes. *Journal of Climate*, 23(1), 97–112. https://doi.org/10.1175/2009jcli3102.1
- D'Odorico, P., Fuentes, J. D., Pockman, W. T., Collins, S. L., He, Y., Medeiros, J. S., et al. (2010). Positive feedback between microclimate and shrub encroachment in the northern Chihuahuan desert. *Ecosphere*, 1(6), art17. https://doi.org/10.1890/es10-00073.1
- Du, J., Fu, Q., Fang, S., Wu, J., He, P., & Quan, Z. (2019). Effects of rapid urbanization on vegetation cover in the metropolises of China over the last four decades. *Ecological Indicators*, 107, 105458. https://doi.org/10.1016/j.ecolind.2019.105458
- Du, J., Shu, J., Yin, J., Yuan, X., Jiaerheng, A., Xiong, S., et al. (2015). Analysis on spatio-temporal trends and drivers in vegetation growth during recent decades in Xinjiang, China. International Journal of Applied Earth Observation and Geoinformation, 38, 216–228. https://doi. org/10.1016/j.jag.2015.01.006
- Duveiller, G., Hooker, J., & Cescatti, A. (2018). The mark of vegetation change on Earth's surface energy balance. *Nature Communications*, 9(1), 679. https://doi.org/10.1038/s41467-017-02810-8
- Fang, J., Piao, S., He, J., & Ma, W. (2004). Increasing terrestrial vegetation activity in China, 1982–1999. Science in China Series C: Life Sciences, 47(3), 229–240. https://doi.org/10.1007/bf03182768
- Fang, J., Yu, G., Liu, L., Hu, S., & Chapin, F. S. (2018). Climate change, human impacts, and carbon sequestration in China. Proceedings of the National Academy of Sciences, 115(16), 4015–4020. https://doi.org/10.1073/pnas.1700304115
- Fettig, C. J., Mortenson, L. A., Bulaon, B. M., & Foulk, P. B. (2019). Tree mortality following drought in the central and southern Sierra Nevada, California, U.S. Forest Ecology and Management, 432, 164–178. https://doi.org/10.1016/j.foreco.2018.09.006
- Fornoff, F., Klein, A.-M., Blüthgen, N., & Staab, M. (2019). Tree diversity increases robustness of multi-trophic interactions. Proceedings of the Royal Society B: Biological Sciences, 286(1898), 20182399. https://doi.org/10.1098/rspb.2018.2399
- Fridley, J. D., Lynn, J. S., Grime, J. P., & Askew, A. P. (2016). Longer growing seasons shift grassland vegetation towards more-productive species. *Nature Climate Change*, 6(9), 865–868. https://doi.org/10.1038/nclimate3032
- Gan, Y. J., Zhang, Y., Kongoli, C., Grassotti, C., Liu, Y. Q., Lee, Y. K., & Seo, D. J. (2021). Evaluation and blending of ATMS and AMSR2 snow water equivalent retrievals over the conterminous United States. *Remote Sensing of Environment*, 254. Go to ISI>://WOS:000609166500001. https://doi.org/10.1016/j.rse.2020.112280
- Gao, M., Piao, S., Chen, A., Yang, H., Liu, Q., Fu, Y. H., & Janssens, I. A. (2019). Divergent changes in the elevational gradient of vegetation activities over the last 30 years. *Nature Communications*, 10(1), 2970. https://doi.org/10.1038/s41467-019-11035-w
- Geng, S., Shi, P., Song, M., Zong, N., Zu, J., & Zhu, W. (2019). Diversity of vegetation composition enhances ecosystem stability along elevational gradients in the Taihang Mountains, China. Ecological Indicators, 104, 594–603. https://doi.org/10.1016/j.ecolind.2019.05.038
- Goodman, D. (1975). The theory of diversity-stability relationships in ecology. The Quarterly Review of Biology, 50(3), 237–266. https://doi.org/10.1086/408563
- Grilli, J., Barabás, G., Michalska-Smith, M. J., & Allesina, S. (2017). Higher-order interactions stabilize dynamics in competitive network models. *Nature*, 548(7666), 210–213.
- Han, X., Zhou, Y., Wang, S., Wang, L., & Hou, Y. (2011). Spatialization approach to 1km grid GDP based on remote sensing. Paper presented at the 2011 international conference on multimedia technology.
- Heise, D. R. (1975). Causal analysis. John Wiley & Sons.
- Hou, X., Liu, S., Cheng, F., Zhang, Y., Dong, S., Su, X., & Liu, G. (2019). Vegetation community composition along disturbance gradients of four typical open-pit mines in Yunnan Province of southwest China. Land Degradation & Development, 30(4), 437–447.

Hu, J.-L., & Wang, S.-C. (2006). Total-factor energy efficiency of regions in China. Energy Policy, 34(17), 3206–3217. https://doi.org/10.1016/j. enpol.2005.06.015

Huang, Y., Chen, Y., Castro-Izaguirre, N., Baruffol, M., Brezzi, M., Lang, A., et al. (2018). Impacts of species richness on productivity in a large-scale subtropical forest experiment. *Science*, 362(6410), 80–83. https://doi.org/10.1126/science.aat6405

Ives, A. R., & Carpenter, S. R. (2007). Stability and diversity of ecosystems. *Science*, 317(5834), 58–62. https://doi.org/10.1126/science.1133258 Jeong, S.-J., Ho, C.-H., Gim, H.-J., & Brown, M. E. (2011). Phenology shifts at start vs. end of growing season in temperate vegetation over the

Northern Hemisphere for the period 1982–2008. Global Change Biology, 17(7), 2385–2399. https://doi.org/10.1111/j.1365-2486.2011.02397.x
Jin, S., Hu, T., Su, Y., Ma, Q., Guan, H., Yang, M., & Guo, Q. (2021). LiVegetation: An investigative tool for vegetation mapping in the era of citizen science. Scientia Sinica Vitae, 51(3), 362–374.

Jorgensen, S. E., & Fath, B. (2014). Encyclopedia of ecology. Newnes.

Ju, J., & Masek, J. G. (2016). The vegetation greenness trend in Canada and US Alaska from 1984–2012 Landsat data. Remote Sensing of Environment, 176, 1–16. https://doi.org/10.1016/j.rse.2016.01.001

Kennedy, C. M., Oakleaf, J. R., Theobald, D. M., Baruch-Mordo, S., & Kiesecker, J. (2019). Managing the middle: A shift in conservation priorities based on the global human modification gradient. *Global Change Biology*, 25(3), 811–826. https://doi.org/10.1111/gcb.14549

- Komatsu, K. J., Avolio, M. L., Lemoine, N. P., Isbell, F., Grman, E., Houseman, G. R., et al. (2019). Global change effects on plant communities are magnified by time and the number of global change factors imposed. *Proceedings of the National Academy of Sciences*, 116(36), 17867–17873. https://doi.org/10.1073/pnas.1819027116
- Körner, C., & Paulsen, J. (2004). A world-wide study of high altitude treeline temperatures. Journal of Biogeography, 31(5), 713–732. https://doi.org/10.1111/j.1365-2699.2003.01043.x
- Lautenbach, S., Jungandreas, A., Blanke, J., Lehsten, V., Mühlner, S., Kühn, I., & Volk, M. (2017). Trade-offs between plant species richness and carbon storage in the context of afforestation – Examples from afforestation scenarios in the Mulde Basin, Germany. *Ecological Indicators*, 73, 139–155. https://doi.org/10.1016/j.ecolind.2016.09.035
- Legates, D. R., & Willmott, C. J. (1990a). Mean seasonal and spatial variability in gauge-corrected, global precipitation. International Journal of Climatology, 10(2), 111–127. https://doi.org/10.1002/joc.3370100202
- Legates, D. R., & Willmott, C. J. (1990b). Mean seasonal and spatial variability in global surface air temperature. *Theoretical and Applied Climatology*, 41(1), 11–21. https://doi.org/10.1007/bf00866198
- Li, C. (2018). Citizen science on the Chinese mainland. In S. Hecker, M. Haklay, A. Bowser, Z. Makuch, J. Vogel, & A. Bonn (Eds.), Citizen science - Innovation in open science, society and policy (pp. 185–189). UCL Press. https://doi.org/10.2307/j.ctv550cf2.19
- Li, X., Jiang, F., Li, L., & Wang, G. (2011). Spatial and temporal variability of precipitation concentration index, concentration degree and concentration period in Xinjiang, China. International Journal of Climatology, 31(11), 1679–1693.
- Li, Y., Piao, S., Li, L. Z. X., Chen, A., Wang, X., Ciais, P., et al. (2018). Divergent hydrological response to large-scale afforestation and vegetation greening in China. Science Advances, 4(5), eaar4182.
- Liu, J., Li, S., Ouyang, Z., Tam, C., & Chen, X. (2008). Ecological and socioeconomic effects of China's policies for ecosystem services. Proceedings of the National Academy of Sciences, 105(28), 9477–9482. https://doi.org/10.1073/pnas.0706436105
- Liu, Y., Li, L., Chen, X., Zhang, R., & Yang, J. (2018). Temporal-spatial variations and influencing factors of vegetation cover in Xinjiang from 1982 to 2013 based on GIMMS-NDVI3g. Global and Planetary Change, 169, 145–155. https://doi.org/10.1016/j.gloplacha.2018.06.005
- Loreau, M., & Mazancourt, C. d. (2013). Biodiversity and ecosystem stability: A synthesis of underlying mechanisms. *Ecology Letters*, 16(s1), 106–115. https://doi.org/10.1111/ele.12073
- Lu, L.-M., Mao, L.-F., Yang, T., Ye, J.-F., Liu, B., Li, H.-L., et al. (2018). Evolutionary history of the angiosperm flora of China. Nature, 554(7691), 234–238. https://doi.org/10.1038/nature25485
- McCann, K. S. (2000). The diversity-stability debate. Nature, 405(6783), 228-233. https://doi.org/10.1038/35012234
- Mougi, A., & Kondoh, M. (2012). Diversity of interaction types and ecological community stability. *Science*, 337(6092), 349–351. https://doi.org/10.1126/science.1220529
- Nagendra, H. (2002). Opposite trends in response for the Shannon and Simpson indices of landscape diversity. Applied Geography, 22(2), 175-186. https://doi.org/10.1016/s0143-6228(02)00002-4
- Peet, R. K. (1975). Relative diversity indices. Ecology, 56(2), 496-498. https://doi.org/10.2307/1934984
- Pennekamp, F., Pontarp, M., Tabi, A., Altermatt, F., Alther, R., Choffat, Y., et al. (2018). Biodiversity increases and decreases ecosystem stability. *Nature*, 563(7729), 109–112. https://doi.org/10.1038/s41586-018-0627-8
- Peters, M. K., Hemp, A., Appelhans, T., Becker, J. N., Behler, C., Classen, A., et al. (2019). Climate–land-use interactions shape tropical mountain biodiversity and ecosystem functions. *Nature*, 568(7750), 88–92. https://doi.org/10.1038/s41586-019-1048-z
- Piao, S., Fang, J., Zhou, L., Guo, Q., Henderson, M., Ji, W., et al. (2003). Interannual variations of monthly and seasonal normalized difference vegetation index (NDVI) in China from 1982 to 1999. Journal of Geophysical Research, 108(D14). https://doi.org/10.1029/2002jd002848
- Piao, S., Yin, G., Tan, J., Cheng, L., Huang, M., Li, Y., et al. (2015). Detection and attribution of vegetation greening trend in China over the last 30 years. *Global Change Biology*, 21(4), 1601–1609. https://doi.org/10.1111/gcb.12795
- Pimm, S. L. (1984). The complexity and stability of ecosystems. *Nature*, 307(5949), 321–326. https://doi.org/10.1038/307321a0
- Pinzon, J. E., & Tucker, C. J. (2014). A non-stationary 1981–2012 AVHRR NDVI3g time series. *Remote Sensing*, 6(8), 6929–6960. https://doi. org/10.3390/rs6086929
- Sankaran, M. (2019). Droughts and the ecological future of tropical savanna vegetation. Journal of Ecology, 107(4), 1531–1549. https://doi.org/10.1111/1365-2745.13195
- Schnabel, F., Schwarz, J. A., Dănescu, A., Fichtner, A., Nock, C. A., Bauhus, J., & Potvin, C. (2019). Drivers of productivity and its temporal stability in a tropical tree diversity experiment. *Global Change Biology*, 25(12), 4257–4272. https://doi.org/10.1111/gcb.14792
- Seto, K. C., Güneralp, B., & Hutyra, L. R. (2012). Global forecasts of urban expansion to 2030 and direct impacts on biodiversity and carbon pools. Proceedings of the National Academy of Sciences, 109(40), 16083–16088. https://doi.org/10.1073/pnas.1211658109
- Shannon, C. E. (1948). A mathematical theory of communication. The Bell System Technical Journal, 27(4), 623–656. https://doi. org/10.1002/j.1538-7305.1948.tb00917.x
- Stokstad, E. (2020). Mountains and monsoons created Tibetan biodiversity. *Science*, *369*(6503), 493. https://doi.org/10.1126/science.369.6503.493 Strassburg, B, B, N., Brooks, T., Feltran-Barbieri, R., Iribarrem, A., Crouzeilles, R., Lovola, R., et al. (2017). Moment of truth for the Cerrado
 - hotspot. Nature Ecology & Evolution, 1(4), 1–3. https://doi.org/10.1038/s41559-017-0099
- Su, Y., Bales, R. C., Ma, Q., Nydick, K., Ray, R. L., Li, W., & Guo, Q. (2017). Emerging stress and relative resiliency of Giant Sequoia groves experiencing multiyear dry periods in a warming climate. *Journal of Geophysical Research: Biogeosciences*, 122(11), 3063–3075. https://doi. org/10.1002/2017jg004005
- Su, Y., Guo, Q., Fry, D. L., Collins, B. M., Kelly, M., Flanagan, J. P., & Battles, J. J. (2016). A vegetation mapping strategy for conifer forests by combining airborne LiDAR data and aerial imagery. *Canadian Journal of Remote Sensing*, 42(1), 1–15. https://doi.org/10.1080/0703899 2.2016.1131114
- Su, Y., Guo, Q., Hu, T., Guan, H., Jin, S., An, S., et al. (2020). An updated vegetation map of China (1:1000000). Science Bulletin, 65(13), 1125–1136. https://doi.org/10.1016/j.scib.2020.04.004
- Tang, X., Zhao, X., Bai, Y., Tang, Z., Wang, W., Zhao, Y., et al. (2018). Carbon pools in China's terrestrial ecosystems: New estimates based on an intensive field survey. Proceedings of the National Academy of Sciences, 115(16), 4021–4026. https://doi.org/10.1073/pnas.1700291115
- Tang, Z., Wang, Z., Zheng, C., & Fang, J. (2006). Biodiversity in China's mountains. Frontiers in Ecology and the Environment, 4(7), 347–352. https://doi.org/10.1890/1540-9295(2006)004[0347:bicm]2.0.co;2
- Tao, S., Fang, J., Zhao, X., Zhao, S., Shen, H., Hu, H., et al. (2015). Rapid loss of lakes on the Mongolian Plateau. Proceedings of the National Academy of Sciences, 112(7), 2281–2286. https://doi.org/10.1073/pnas.1411748112
- Thomson, E., Malhi, Y., Bartholomeus, H., Oliveras, I., Gvozdevaite, A., Peprah, T., et al. (2018). Mapping the leaf economic spectrum across West African tropical forests using UAV-acquired hyperspectral imagery. *Remote Sensing*, 10(10), 1532. https://doi.org/10.3390/rs10101532
- Tian, F., Herzschuh, U., Mischke, S., & Schlütz, F. (2014). What drives the recent intensified vegetation degradation in Mongolia Climate change or human activity? *The Holocene*, 24(10), 1206–1215. https://doi.org/10.1177/0959683614540958



Tucker, C. J., Pinzon, J. E., Brown, M. E., Slayback, D. A., Pak, E. W., Mahoney, R., et al. (2005). An extended AVHRR 8-km NDVI dataset compatible with MODIS and SPOT vegetation NDVI data. *International Journal of Remote Sensing*, 26(20), 4485–4498. https://doi. org/10.1080/01431160500168686

van Leeuwen, W. J. D., Huete, A. R., & Laing, T. W. (1999). MODIS vegetation index compositing approach: A prototype with AVHRR data. Remote Sensing of Environment, 69(3), 264–280. https://doi.org/10.1016/s0034-4257(99)00022-x

- Wang, Y., Ziv, G., Adami, M., Almeida, C. A. d., Antunes, J. F. G., Coutinho, A. C., et al. (2020). Upturn in secondary forest clearing buffers primary forest loss in the Brazilian Amazon. *Nature Sustainability*, 3(4), 290–295. https://doi.org/10.1038/s41893-019-0470-4
- Wang, Z., Deng, X., Song, W., Li, Z., & Chen, J. (2017). What is the main cause of grassland degradation? A case study of grassland ecosystem service in the middle-south Inner Mongolia. *Catena*, 150, 100–107. https://doi.org/10.1016/j.catena.2016.11.014

Wang, Z., Fang, J., Tang, Z., & Lin, X. (2012). Relative role of contemporary environment versus history in shaping diversity patterns of China's woody plants. *Ecography*, 35(12), 1124–1133. https://doi.org/10.1111/j.1600-0587.2011.06781.x

Willmott, C. J. M. K. (2001). Terrestrial air temperature and precipitation: Monthly and annual time series (1950–1999). Retrieved from http:// climate.geog.udel.edu/~climate/html_pages/README.ghcn_ts2.html

Wu, Z. Y. W. (1985). China's physical geography: Phytogeography. Science Press.

Zhang, D., Fengquan, L., & Jianmin, B. (2000). Eco-environmental effects of the Qinghai-Tibet plateau uplift during the Quaternary in China. Environmental Geology, 39, 1352–1358.

Zhang, J.-T., Zhang, M., & Mian, R. (2016). Effects of elevation and disturbance gradients on forest diversity in the Wulingshan Nature Reserve, North China. Environmental Earth Sciences, 75(10), 904. https://doi.org/10.1007/s12665-016-5734-y

- Zhang, X., Sun, S., Yong, S., Zhou, Z., & Wang, R. (2007). Vegetation map of the People's Republic of China (1: 1000000). Geology Publishing House.
- Zhao, X., Su, Y., Hu, T., Chen, L., Gao, S., Wang, R., et al. (2018). A global corrected SRTM DEM product for vegetated areas. *Remote Sensing Letters*, 9(4), 393–402. https://doi.org/10.1080/2150704x.2018.1425560
- Zhu, Z., Piao, S., Myneni, R. B., Huang, M., Zeng, Z., Canadell, J. G., et al. (2016). Greening of the Earth and its drivers. *Nature Climate Change*, 6(8), 791–795. https://doi.org/10.1038/nclimate3004